Author's response to reviews

Title: Vitamin A deficiency during pregnancy of HIV infected and non-infected women in tropical settings of Northwest Ethiopia

Authors:

Andargachew Mulu (andargachewmulu@yahoo.com)
Afework Kassu (afeworkkkassu@yahoo.com)
Kahsay Huruy (kasaye88@yahoo.com)
Birhanemeskel Tegene (birehane89@yahoo.com)
Gashaw Yitayaw (gashiy55@yahoo.com)
Masayo Nakamori (masana97@hotmail.com)
Nhien Nguyen (nhiennguyenvan@yahoo.com)
Asegede Bekele (assba1add@yahoo.co.uk)
Yared Wondimihun (ywondimkun@yahoo.com)
Shigeru Yamamoto (yamamoto.shigeru@ocha.ac.jp)
Fusao Ota (ota-fusao@me.pikara.ne.jp)

Version: 5 Date: 14 July 2011

Author's response to reviews:

Dear the Editorial Office members,

We made the following changes:

Major revisions: None

Affiliations: Each affiliation are rewritten as indicated (written in full following format: Department, Institute, (Street), City, (Postal code), Country. Author’ contributions: Please state that all authors read and approved the final manuscript.

Author’s contribution: All authors have read and approved the final version of the manuscript.

Tables: Tables 2 and 3 are cited. Tables 1, 2, and 4 are formatted using MS Word auto-format tool and created in the same manner as Table 3.

Minor revisions: Highlight and tracks are removed.

Introduction: renamed as 'Background'.

Typography: Typographical errors are corrected.

References: References are cited in numerical order.

Lastly, I declare that all the authors have read and approved the final manuscript.